Author’s response to reviews

Title: The role of MMP-12 gene polymorphism -82 A-to-G (rs2276109) in immunopathology of COPD in Polish patients: a case control study

Authors:
Iwona Gilowska (i.gilowska@po.opole.pl)
Edyta Majorczyk (e.majorczyk@po.opole.pl)
Łukasz Kasper (kasper@mp.pl)
Katarzyna Bogacz (k.bogacz@po.opole.pl)
Jan Szczegielniak (j.szczegielniak@po.opole.pl)
Marta Kasper (marta.1.kasper@uj.edu.pl)
Jacek Kaczmarski (j.kaczmarski@po.opole.pl)
Aleksandra Skomudek (a.skomudek@po.opole.pl)
Marcin Czerwinski (m.czerwinski@po.opole.pl)
Krzysztof Sładek (mmsladek@cyf-kr.edu.pl)

Version: 1 Date: 10 Oct 2018

Author’s response to reviews:

Dear Dr. Pink An,

In accordance to Editor's comments, I am submitting a revised version of our manuscript.

L177: HWE P>0.05? yes, p value is non-significant and > 0.05, a correction has been done
L374-374: #, groups for CNV and protein

374 levels analyses, please define CNV - CNV abbreviation was redundant and was removed
Table 2: put OR (CI) into the table. - ORs (CI) for G allele were placed into the Table 2.
Thank you for your comments.

Yours sincerely,

Edyta Majerczyk